cs330-01 & 91: Discrete Structures Spring 1996

Computer Science and Applied Mathematics

Programming Assignment 2 - Grading

In the following, p1 denotes the first problem of the assignment (scientific notation for numbers) and p2 denotes the second problem (the security device). RE stands for Regular Expression.

Student:

Functionality pass fail Readability pass fail

p1 accepts 3.1415e0 -0.2 Memo included -1 p1 accepts +3.1415e0 -0.2 Standard memo format -0.1 p1 accepts -3.1415e0 -0.2 RE for p1 included in memo -0.5 p1 accepts 3.1415e+0 -0.2 RE for p2 included in memo -0.5 p1 accepts 3.1415e-0 -0.2 Security code included in memo -0.1 p1 accepts +3.1415E-0 -0.2 FA diagram for p1 included -0.5 p1 accepts -3.1415E+0 -0.2 FA diagram for p2 included -0.5 p1 rejects 3.1415 -0.2 README file present for p1 -0.5 p1 rejects 0.31415e1 -0.2 Proper README for p1 -0.5 p1 rejects 31.415e-1 -0.2 File level documentation for p1 -0.5 p1 rejects +-3.1415e0 -0.2 Function level documentation for p1 -0.5 p1 rejects 3..1415e0 -0.2 Well commented code for p1 -0.5 p1 rejects 3.1415e0.0 -0.2 README file present for p2 -0.5 p1 rejects 3.1.415e0 -0.2 Proper README for p2 -0.5 p1 rejects empty input -0.2 File level documentation for p2 -0.5 p1 rejects single letter input -0.2 Function level documentation for p2 -0.5 Correct security code for p2 -1 Well commented code for p2 -0.5 p2 unlocks properly -1 p2 locks properly -1 Correct estimate for breaking lock +1 Actual measurements for breaking +1

Functionality x Readability = Final mark